%% Created by Maple V Release 4 (SUN SPARC SOLARIS)
%% Source Worksheet: graph.mws
%% Generated: Mon Nov  9 08:45:24 1998
\documentclass{article}
\usepackage{maple2e}
\DefineParaStyle{Author}
\DefineParaStyle{Maple Plot}
\DefineParaStyle{Title}
\DefineParaStyle{Warning}
\begin{document}
\begin{maplegroup}
\begin{mapleinput}
\end{mapleinput}

\mapleresult
\begin{maplettyout}
Warning, new definition for norm
\end{maplettyout}

\begin{maplettyout}
Warning, new definition for trace
\end{maplettyout}

\end{maplegroup}
\begin{maplegroup}
\begin{Title}
Parametric Plot of Solution
\end{Title}

\begin{Author}
for problem 7.1.7(a), Set \# 20
\end{Author}

\mapleinline{inert}{1d}{plot([2.5*exp(-t)-.5*exp(-3*t),2.5*exp(-t)+.5*
exp(-3*t),t=0..20]);}{%
}

\mapleresult
\begin{center}
\mapleplot{graph01.eps}
\end{center}

\end{maplegroup}
\begin{maplegroup}
\begin{Title}
 Parametric Plot of Solution
\end{Title}

\begin{mapleinput}
\end{mapleinput}

\end{maplegroup}
\end{document}
%% End of Maple V Output
